A ONE HEALTH APPROACH TO FOOD SAFETY AND THE RIGHT TO HEALTH IN A GLOBAL GOVERNANCE

NAIARA ARRIOLA ECHANIZ¹

Abstract: The global regulation of food safety remains remarkably limited, despite growing recognition of their transboundary implications. This article examines how a cosmopolitan approach to global governance can strengthen the protection of the right to health by integrating the One Health paradigm into food safety regulation. It argues that food safety should not be treated as a merely technical issue, but as a constitutional and human rights concern within a globally interdependent context. Through the analysis of selected WTO disputes, the article illustrates how tensions between trade obligations and human rights protections reveal deeper normative and institutional gaps in global food governance.

Keywords: Right to health, Food safety, Cosmopolitanism, One Health, ESG, Sustainable Development Goals, Global Governance.

SUMMARY: 1. Introduction. 2. Cosmopolitanism and Global Human Rights Governance. 3. The One Health Paradigm: A Normative Approach to Interdependent Rights. 4. Case Study: WTO-EU Food Safety Tensions as Human Rights Dilemmas. 5. Food Safety and ESG: Toward a Rights-Based Frame-Work. 6. Conclusions.

1. Introduction

In an increasingly interconnected world, the distinction between national and global health risks has become blurred. This is particularly evident in the context of food safety, where the emergence and spread of foodborne illnesses, zoonotic diseases and antimicrobial resistance exemplify contemporary public health challenges that transcend national borders. These developments necessitate legal and policy frameworks that transcend national borders and address health risks comprehensively and integrally. Against this backdrop, the One Health paradigm has gained traction as a legal and operational approach that recognises the interconnectedness of human health, animal health, and the environment.

At the same time, the right to health has evolved from a state-centered entitlement to a global concern that requires coordinated responses beyond domestic policies. As pandemics and food crises have shown, the enjoyment of health rights in one part of the world is contingent upon the existence of robust health and food regulations elsewhere. Consequently, the global governance of health and food safety cannot be reduced to an isolated national policy. In fact, the right to health (and food safety) must be understood as a normative enterprise that requires the protection of justice and accountability at a global level.

¹ Professor of Constitutional Law and WTO Law. Universidad Pontificia Comillas-ICADE. Madrid, Spain (narriola@comillas.edu).

Why is the right to health, specifically food safety, not regulated on a global level? This article argues that a cosmopolitan approach to global governance offers a valuable framework to rethink food safety regulation globally, through the lens of the right to health. Cosmopolitanism scholars assert that all individuals, regardless of nationality or citizenship, are entitled to equal moral and legal protection. In the field of global health law, this perspective entails regulatory standards reflect shared human interests and interdependent vulnerabilities. In this sense, cosmopolitanism provides a conceptual intersection between the universality of human rights and the systemic logic of One Health.

The definition of One Health that suits this paper is the one proposed by the One Health High- Level Expert Panel. "One Health is an integrated, unifying approach that aims to sustainably balance and optimize the health of humans, animals, plants and ecosystems. It recognizes the health of humans, domestic and wild animals, plants and the wider environment (including ecosystems) are closely linked and interdependent" (One Health High-Level Expert Panel *et al.*, 2022, p. 4). While the One Health paradigm offers an essential integrative framework for addressing the complex interconnections between human, animal, and environmental health, it remains largely embedded in soft law instruments, voluntary cooperation, and institutional declarations. This limits its capacity to impose binding obligations on states or to guarantee accountability in the governance of global health risks. In this context, the right to health, as enshrined in international human rights law - notably Article 12 of the International Covenant on Economic, Social and Cultural Rights – provides the normative foundation that could strengthen and consolidate the aspirations of One Health.

In this sense, the Office of the United Nations High Commissioner for Human Rights has affirmed that food safety is a fundamental component of the right to adequate food and an essential requirement for realising the right to health. General Comment No. 14 of the UN Committee on Economic, Social and Cultural Rights reinforces this interpretation by clarifying that states are obliged to ensure safe food systems and environmental health as part of their human rights commitments. Yet despite its growing normative relevance, One Health remains embedded in soft law, raising the urgent question of whether it can serve as a vehicle for integrated global governance.

The article explores how food safety regulation can be aligned with cosmopolitan and One Health principles in practical terms (case study). Besides through the integration of Environmental, Social, and Governance frameworks and the Sustainable Development Goals. ESG frameworks, although primarily rooted in corporate governance and financial accountability, have increasingly influenced global discussions on food system sustainability and corporate responsibility in ensuring public health and food safety. The Sustainable Development Goals (SDGs), for their part, embody an international consensus around shared objectives that require multi-level governance, long-term vision, and cross-sectoral coordination. Nonetheless, the proposal is not just a matter of compliance. The case study of WTO-EU food safety analyses the tensions as human rights dilemmas and will give us examples of how to (or not to) build a solid policy reform.

To ground the theoretical and normative discussion in real-world dynamics, the article presents a case study of selected WTO-EU food safety paradigmatic disputes:

EC – Hormones, EC – Biotech and EC – Seal Products. These disputes, along with the recent controversy surrounding pesticide-related trade measures, highlight the legal and political tensions between trade liberalisation and public health protection. Each case demonstrates how food safety measures that are presented as human rights or environmental safeguards may conflict with the WTO's rules- based system, particularly when ethical or precautionary justifications are employed rather than strict scientific consensus. These disputes demonstrate the regulatory and power imbalances at the core of global food governance, offering a valuable perspective through which to evaluate the feasibility of incorporating One Health and the right to health into a more comprehensive international legal framework.

This article is divided into five sections. Following the introduction, Section 2 outlines the foundations of cosmopolitanism as a legal and political theory of global justice and its implications for human rights governance, with a particular focus on the right to health and food safety. Section 3 explores the concept of One Health, examining its scientific origins, institutional development and evolving role in global health policy. Section 4 examines the part played by ESG frameworks and the SDGs in promoting a rights-based approach to food safety, considering how these standards either contribute to or fall short of a cosmopolitan One Health vision. Section 5 presents a case study of WTO–EU food safety tensions as human rights dilemmas. It examines how the EU positions its regulatory standards and the extent to which these embody the principles of One Health and cosmopolitanism. The conclusion reflects on the implications of adopting a One Health approach to food safety and the right to health, proposing directions for policy reform. By integrating these perspectives, the article aims to contribute to the development of a more coherent, inclusive and ethically grounded legal framework for global food governance.

In the final section, which is dedicated to conclusions, the article argues that global food safety governance must move beyond its current fragmented, trade-centred approach and embrace a more coherent, inclusive, rights-based framework. Adopting the One Health approach, applying cosmopolitan legal theory and incorporating emerging Environmental, Social and Governance (ESG) mechanisms could shift the focus of global governance towards safeguarding food safety as a fundamental human right. The case study of WTO–EU disputes illustrates the inadequacy of the current trade regime for safeguarding health and ecological integrity, thereby reinforcing the need for normative and institutional reform. In response, the article sets out key policy recommendations for enshrining the right to health and the One Health approach in international law. This would advance a cosmopolitan vision of global food governance that prioritises health, sustainability, and justice.

2. Cosmopolitanism and Global Human Rights Governance

Why is the right to health, specifically food safety, not regulated on a global level? Historically, health policy has been considered an integral part of national sovereignty, enabling states to exercise considerable autonomy and resist the imposition of binding international obligations (Fidler, 2004; Lange, 2021). States have traditionally

viewed health policy as an essential aspect of their autonomy and have resisted any international obligations that might restrict their regulatory discretion. The limited global regulation of the right to health, particularly with regard to food safety, can be traced back to this primacy.

In this context, the international legal framework for food safety remains highly fragmented. Guidance on this issue is scattered across different institutional frameworks, primarily the Codex Alimentarius Commission (set up by the World Health Organization and the Food and Agriculture Organization) and the World Trade Organization's Agreement on the Application of Sanitary and Phytosanitary Measures (SPS Agreement) (Fortin, 2023). However, these two instruments operate according to different regulatory principles. The first is based on scientific harmonisation, while the second follows trade facilitation. Neither has a unified, rights-based approach that can effectively protect food safety as an integral component of the right to health under international human rights law (Arcuri, 2015). Consequently, the international legal landscape for food safety is fragmented, with guidance dispersed among institutions such as the Codex Alimentarius and the WTO's SPS Agreement. These institutions operate under different logics and lack a unified, rights-based approach (Petersmann, 2017). This highlights the challenge of incorporating human rights into global governance structures.

Cosmopolitanism as a Theoretical Framework and Starting Point. The historical precedents of cosmopolitan theory can be traced back to Immanuel Kant and his proposal for a federation of democratic states to ensure peace and grant individuals the right to world citizenship (Dudley and Engelhard, 2011). Similar ideas can be found in the work of Hans Kelsen, who viewed international law as superior to national systems and proposed the establishment of an international court to promote peace. However, he was cautious about the formation of a world state (*Civitas Maxima*). The biggest obstacle was, and arguably still is, the national sovereignty of states. Moreover, this is a political rather than a legal question (Kelsen, 2001).

Cosmopolitanism proposes the existence of supranational institutions over and above national state institutions. It begins with the supposed crisis of the nation state, which has seen its effectiveness and capacity to act in many areas diminish due to globalisation and global interconnectedness (Arriola Echaniz, 2008; 2017; 2019). Cosmopolitanism offers a vision that transcends the nation-state, advocating the global institutionalisation of democracy and human rights. It is rooted in the idea that everyone is a member of a global community and deserves equal moral consideration and protection, regardless of their nationality (Moka-Mubelo, 2017). This perspective challenges the traditional, state-centric model by emphasising the importance of universal human rights and shared responsibility for protecting them. Rather than being opposed to globalisation, cosmopolitanism seeks to humanise it by prioritising social welfare, sustainable development and cultural respect (Held, 2010). According to Beck (2002; 2005), cosmopolitanism is a democratic project that aims to extend democracy into the transnational sphere.

Cosmopolitan theory is proposed as an alternative to the nation state. It advocates the institutionalisation of democracy at a global level, focusing on ensuring that the social

and political rights of all citizens worldwide are respected. It seeks to re-politicise the new power structures that operate beyond the nation state, and is not opposed to globalisation *per se*, but rather to hegemonic neoliberal globalisation. In fact, it advocates a humanised globalisation that prioritises social welfare, sustainable development, cultural respect, and the protection of human rights (Arriola Echaniz, 2019).

Fundamental Cosmopolitan Principles for Law. Cosmopolitan principles form the foundation for acknowledging human rights on a global scale. Firstly, all human beings are recognised as having the capacity for active action or self-determination. This capacity is accompanied by a responsibility to prevent and repair damage and to respect sustainability at the global level. The second principle recognises the individual responsibility for the consequences of actions derived from recognised rights (Arriola Echaniz, 2019).

Fundamental rights must be respected and guaranteed to all people at the global level, regardless of their nationality or state of residence. There has been discussion about a new global law that would supersede international law by operating under the principle of personality rather than territoriality and providing a guiding ethical basis for global social democracy (Held, 2016). The belief that they are the exclusive property of Western culture must be overcome. Cosmopolitan authors argue that applying them helps develop and encourage participation in global civil society. This participation should be encouraged through a non-coercive political decision-making process based on consent, which would overcome the democratic deficit of international organisations (Arriola Echaniz, 2019).

Protecting Rights on a Global Scale. The application of the cosmopolitan doctrine as a legal concept is primarily evident in the suggestion that, where national legal systems are dysfunctional, we should rely on new supranational sources of legitimacy. The idea is to move towards transnational legislation that transcends the legal systems of national states (Arriola Echaniz, 2008).

The main path from 'top-down' cosmopolitanism is legal and institutional development driven by traditional actors such as states and international organisations. This involves establishing a higher, supranational legal order. The democratic cosmopolitan state is based on cosmopolitan democratic law, which is a form of public law with a democratic character in the transnational space that transcends national interests. The main objective of this law is to guarantee civil liberties and cultural diversity at all levels (local, regional, national and global). This requires the cosmopolitanisation of law. Sovereignty would become shared and inclusive, overcoming its territorial and indivisible nature (Arriola Echaniz, 2008; 2019).

In the case of 'cosmopolitanism from below', while the principles are similar, institutionalisation would be driven by social mobilisation (Arriola Echaniz, 2019). Boaventura de Sousa Santos speaks of counter-hegemonic globalisation and subaltern cosmopolitan legality (2005), seeking socio-legal approaches that clarify hegemonic institutional changes, document resistance and affirm the potential for subversion, while imagining legal institutions built 'from below'. This involves directing analysis towards

legal alternatives created by grassroots movements (De Sousa Santos and Rodríguez Garavito, 2007). Richard Falk (2002) proposes normative democracy as a unifying force of global civil society that goes beyond elections and includes additional guarantees to ensure that the exercise of power promotes human welfare and ecological sustainability and gives citizens access to decision-making arenas.

Key cosmopolitan principles include the recognition of every person's capacity for self- determination and the responsibility to prevent harm and promote sustainability globally. These principles underpin the call for a global legal order that guarantees fundamental rights to all, moving beyond the limitations of territoriality. However, critics argue that extending rights beyond the state without robust accountability mechanisms risks creating dependency and undermining existing democratic rights. The debate continues over how to balance respect for state sovereignty with the need for effective global human rights protection (Cabrera, 2020).

Constitutionalizing the globalisation. The notion of 'humanising' international organisations and international law itself reflects the idea of protection and protection of rights on a global scale. This process aims to reinforce the protection and respect of individuals' rights at an international level. Human rights are asserted at the supranational level to discipline the international market, for example (Arriola Echaniz, 2017). From this perspective, globalisation undergoes a process of constitutionalisation, developing institutions for the protection of human rights beyond the state (Arriola Echaniz, 2017).

Supranational courts such as the European Court of Human Rights and the Inter-American Court of Human Rights exemplify efforts to protect rights beyond national jurisdictions, though enforcement remains uneven (Rozpedowski, 2019). A paradigmatic example of supranational protection is the European system, comprising the European Court of Human Rights and the Court of Justice of the European Union. These courts advance the protection of rights beyond the remit of national constitutional courts. The Inter-American system, comprising the Inter-American Court of Human Rights, is also developing such protection. However, it encounters challenges in enforcing judgments due to the varying ratifications of the Pact of San José among American states.

As we will see in the case study, the analysis in question debates whether institutions such as the World Trade Organisation (WTO) fulfil 'constitutional functions' by guaranteeing economic freedoms and incorporating non-trade values such as the environment and human rights into their dispute settlement system and law (Petersmann, 2008). This is considered a 'constitutionalisation' of WTO law, and consequently of international law (Trachtman, 2014). Some scholars advocate for the 'constitutionalisation' of global institutions like the WTO, integrating human rights and social justice into their frameworks to address normative gaps in global governance (Petersmann, 2017). Comparative analyses suggest that cosmopolitan international agreements empowering citizens and decentralized compliance procedures are more effective than traditional state-centric agreements.

Recent scholarship highlights that the constitutionalization of international law should not only reinforce individual rights but also address the social dimension of global

governance. In fact, the legitimacy of supranational legal orders, therefore, increasingly depends on their capacity to protect not only civil and political rights but also social rights, as emphasized by Peters in her call for a 'social limb' in global constitutionalism (Peters, 2018). This approach mitigates the neoliberal bias of existing frameworks and supports the development of a more inclusive and equitable international legal order.

Final considerations. Despite its appeal, cosmopolitanism faces practical challenges, including the absence of enforceable global mechanisms and the risk of imposing Western values (Cabrera, 2020; Dahrendorf, 2005; Zolo, 2000). Nevertheless, it continues to inspire reforms that aim to reconcile national sovereignty with the protection of universal human rights. This change must be both top-down and bottom-up. It must be both institutional and social (Held, 2010). While cosmopolitanism provides a compelling ethical and legal framework for advancing global human rights governance, realising this vision requires overcoming significant political, legal, and cultural obstacles. The ongoing evolution of supranational institutions and the push to integrate human rights more fully into global governance demonstrate the continued importance of, and difficulties associated with, cosmopolitan ideals.

This tension is more than clear in the regulation of food safety within the framework of the World Trade Organization. Despite the growing recognition of the global nature of health and food safety issues, the international legal framework continues to reflect a normative separation between trade and human rights. The SPS Agreement, which governs national food safety measures within the multilateral trading system, prioritises non-discrimination and scientific justification. There is no reference in there to human rights obligations. This institutional and legal design perpetuates the idea that food safety is primarily an issue of market access rather than a matter of public health or social justice. Furthermore, the absence of a binding global mechanism to enforce economic and social rights, such as the right to health or the right to safe food, creates a normative gap in global governance. Legal scholars such as Ernst-Ulrich Petersmann (2012, 2013) have long criticised this discrepancy. They have argued for the integration of human rights into WTO law and for reconsidering trade policy as a means of advancing the constitutional principles of justice and sustainability. Section 4, which contains case studies of WTO-EU food safety disputes, will demonstrate these normative tensions and provide an introduction to the examination of how cosmopolitan principles could transform global trade law into a more rights-based and ethically coherent framework.

3. THE ONE HEALTH PARADIGM: A NORMATIVE APPROACH TO INTERDEPENDENT RIGHTS

Preliminary considerations. In an era where transboundary health risks are a major concern, the traditional approach of national legal regulation is no longer sufficient. One Health has emerged as a response to this reality, offering an integrative framework that acknowledges the systemic interconnection of biological, ecological, and social systems. This approach is of pertinence to the governance of food safety in a globalised world, where zoonotic diseases, antimicrobial resistance, and environmental degradation increasingly challenge the effectiveness of isolated and sectoral policy responses. Of

particular significance is the way in which the principles underlying One Health are grounded in international human rights law, primarily through the recognition of the right to health. According to this research's topic, the One Health approach aims to achieve global food safety regulation by promoting transdisciplinary collaboration across sectors and disciplines. This is intended to overcome regulatory silos and foster holistic responses to complex health threats. Article 12 of the International Covenant on Economic, Social and Cultural Rights formally recognises the inherent right of all individuals to the highest attainable standard of physical and mental health. This right encompasses access to essential resources such as safe and nutritious food, clean water, and a healthy environment.

In this sense, the Office of the United Nations High Commissioner for Human Rights has observed that food safety is a fundamental component of the right to adequate food and an essential requirement for achieving the right to health. This interpretation has been further developed in General Comment No. 14 of the UN Committee on Economic, Social and Cultural Rights. The document clarifies that states have an obligation to respect, protect and fulfil the conditions necessary for safe food systems and environmental health as part of the right to health (Hunt, 2016). However, despite its growing prominence in global health discourse, the One Health approach remains rooted in soft law instruments, institutional declarations and voluntary cooperation. This means it lacks the binding legal obligations that could consolidate its normative aspirations. Against this backdrop, an urgent question arises: Is One Health a vehicle to move from fragmented soft law to integrated global governance?

One Health Paradigm: as a concept. The One Health approach is guided by a holistic understanding of the interface between humans, animals and the environment, and emphasises the importance of interdisciplinary collaboration in effectively tackling global health issues. Its primary goal is to enhance global health equity and mitigate health disparities by addressing critical global health concerns, including zoonotic diseases, antimicrobial resistance, food security, and climate change (Zhang et al., 2024, p. 2). Prominent international organisations, collectively known as the Quadripartite, have formally defined the One Health approach and play a key role in promoting its global implementation. These organisations are the Food and Agriculture Organization of the United Nations, the World Organisation for Animal Health, the World Health Organization and the United Nations Environment Programme. The One Health High-Level Expert Panel also provides implementation guidance (Zhou et al., 2025). In order to foster global capacity building, enhance preparedness for preventing, predicting, identifying and responding to these complex health challenges, and promote sustainable development, the Quadripartite launched the One Health Joint Plan of Action in October 2022 (Zhang et al., 2024, p. 2).

Global One Health Index Report 2022. The 2022 Global One Health Index Report, primarily developed by Zhou *et al.* (2025) as an independent academic tool, presents the latest findings of the Global One Health Index (GOHI). GOHI is an innovative evaluation tool designed to assess and promote the implementation of the One Health approach worldwide. This open-access publication emphasises the development and importance of the GOHI in monitoring One Health practices in 160 countries and

territories. It details country-level scores and provides key insights into global and regional performance. The report also offers tailored policy recommendations aimed at strengthening the adoption of One Health principles by United Nations agencies, national governments and local communities. Building on a 2021 pilot study, the report is a valuable resource for researchers, practitioners, policymakers and other stakeholders seeking to advance One Health integration in public health, environmental management and animal health governance. The following analysis examines the report's structure, findings and normative implications for global health governance.

From reading the 'Global One Health Index Report 2022' (Zhou *et al.*, 2025), we can extract the following key aspects of the One Health paradigm: interconnectedness; addressing complex health challenges; multi-sectoral collaboration; and integration into governance. First, Interconnectedness. At its core, One Health is based on the principle that the health of humans, animals, and the environment are inextricably linked. Human activities directly impact the environment, which in turn affects the health of animals and, ultimately, human well-being.

Secondly, the need for an integrated approach has become increasingly evident due to the emergence of significant public health threats at the human-animal-environment interface, i.e. complex health challenges. One Health is crucial for addressing multifaceted issues such as zoonotic disease outbreaks. These outbreaks are responsible for a significant proportion of new and re-emerging infectious diseases. Additionally, antimicrobial resistance is a complex challenge that is exacerbated by interactions across sectors, and climate change directly impacts health outcomes across all three domains. Food insecurity is another critical area where an integrative perspective is needed.

Thirdly, the One Health approach emphasises the vital need for interdisciplinary and cross-border collaboration between different sectors and disciplines, as well as at community level. The aim is to remove existing obstacles in order to develop a comprehensive One Health framework geared towards societal action. This involves placing a stronger focus on improving communication, coordination, collaboration and capacity building within national development strategies and international cooperation efforts.

Fourthly, the strategic promotion of the One Health concept involves integrating it into global governance and establishing it as a high-level agenda item. This requires engaging a wide array of stakeholders, such as international, non-governmental and civil society organisations, as well as the private sector and academia, to ensure the sustainability of the cooperation network. Intergovernmental dialogue is also encouraged in order to establish long-term financing mechanisms, global early warning and response systems, and preparedness for future pandemics in collaboration with international partners.

One Health Joint Plan of Action. The One Health Joint Plan of Action (OHJPA) for 2022-2026 represents a significant normative and strategic milestone in advancing the One Health paradigm. Developed and adopted by the Quadripartite, the OHJPA aims to provide a unified operational framework for the global implementation of One

Health principles. Its primary objective is to coordinate international efforts, set priorities and guide policy action, institutional collaboration and capacity building at the global, regional and national levels. By offering a structured approach to tackling complex health challenges at the interface between humans, animals, and the environment, the OHJPA aims to transform the One Health concept from an aspirational framework into a coherent model for integrated global health governance (World Health Organization *et al.*, 2022). This analysis examines the content, strategic priorities and normative significance of the OHJPA within the broader context of One Health implementation.

This plan was created in response to international calls for preventing future pandemics and promoting sustainable health through the One Health approach and acknowledges that economic development has often occurred at the expense of ecosystems and environmental health, leading to tremendous pressures on natural systems (World Health Organization *et al.*, 2022, p. 3). It emphasizes the urgent need to reassess and transform these interactions to achieve economic, environmental, and social sustainability, thereby contributing to the Sustainable Development Goals (World Health Organization *et al.*, 2022, pp. 22-26).

The core of the plan outlines Action tracks (World Health Organization *et al.*, 2022, Part. 3, pp. 21-55). Regarding the core of this article, we would dive into the Action track 4: Strengthening the assessment, management and communication of food safety risks. Action Track 4 is designed to create a more resilient and effective global food safety system by embedding the holistic One Health philosophy into policy, data collection, risk assessment, and surveillance, acknowledging the complex interplay between all living organisms and their shared environment (World Health Organization *et al.*, 2022, pp. 39-43).

Food systems, encompassing the entire pathway from production to consumption, are situated at the intersection of human, animal, plant, and environmental health. The methods employed in food production influence not only the safety of the final product, but also animal health and welfare, plant health, and environmental integrity. Conversely, the conditions of the production environment, as well as the health of animals and plants, can significantly affect food safety. This intricate interdependence underscores the necessity of adopting a One Health approach to effectively safeguard food safety, while also highlighting that ensuring food safety is integral to advancing One Health objectives (World Health Organization *et al.*, 2022, pp. 39-40).

This action aims to bolster national food safety systems by focusing on the establishment or improvement of critical infrastructure components. This includes food safety legislation, standards, guidelines, laboratory capacity, food control activities, and emergency preparedness and response. The goal is to help countries reduce risks from unsafe food, ensure food authenticity, and enhance fair and safe trade, with particular attention to low- and middle-income countries (World Health Organization *et al.*, 2022, pp. 40-41). Action Track 4 is structured into three interdependent actions, each with specific activities, deliverables, and timelines: Action 4.1: Strengthening the One Health approach in national food control systems and food safety coordination; Action 4.2: Utilize and improve food systems data and analysis, scientific evidence and risk assessment in

developing policy and making integrated risk management decisions; and Action 4.3: Foster the adoption of the One Health approach in national foodborne disease surveillance systems and research for the detection and monitoring of foodborne disease and food contamination.

Final considerations: the right to food safety as a human right. Although the One Health paradigm offers a vital integrative framework for addressing the complex relationships between human, animal and environmental health, it is primarily based on non-binding legal instruments, voluntary cooperation and institutional declarations. Integrating One Health within a rights-based framework is crucial for transforming its holistic vision into enforceable commitments and coherent global governance.

There are clear examples of how the right to health creates legal obligations for states. The right to health obliges states to respect, protect and fulfil the conditions necessary for health, including food and environmental safety (CESCR, 2000). For instance, states must regulate private actors to ensure food safety, adopt policies to combat antimicrobial resistance and provide access to clean water and safe nutrition (CESCR, 2000, paras. 33-37).

Ensuring food safety on a global scale requires scientific collaboration, as well as regulatory mechanisms that can monitor compliance, enforce standards, and apply sanctions. However, most national states have been reluctant to cede these capacities to international bodies. This reluctance reflects deeper conflicts between national sovereignty, economic competitiveness, and the often contested neutrality of technological standards. As Jasanoff (2004) has argued, the role of science in policymaking is often influenced by political and economic agendas, determining which risks are prioritised and which knowledge is given legitimacy. Similarly, De Schutter (2014) emphasises that food governance cannot be reduced to technical efficiency or market logic. On the contrary it must be anchored in rights-based principles that address power asymmetries and ensure accountability. These insights illuminate the complex landscape in which ESG frameworks are now emerging as influential tools for advancing food safety. This raises the question of whether they can meaningfully contribute to a more rights-oriented and globally coherent system of food governance.

Against this backdrop, ESG frameworks have gained prominence as instruments capable of shaping corporate behaviour and guiding investment decisions towards more sustainable and socially responsible practices². Originally developed in financial and business contexts, ESG criteria are now being promoted by international organisations, investors, and regulators as a means of achieving long-term sustainability goals than can be related with food safety, environmental health, and labour rights. Thanks to their flexibility and global reach, ESG criteria have the potential to generate accountability through non-state mechanisms such as market access, reputational pressure and stakeholder reporting. However, this potential has limitations. ESG standards remain

_

² In 2004, the term 'ESG' became official after its first mainstream appearance in a report titled, "Who Cares Wins" (International Finance Corporation, 2004).

largely voluntary, inconsistent and susceptible to corporate influence. The challenge for this article, then, is to assess how ESG can evolve from a market-oriented compliance tool into a normative framework that enforces human rights obligations and supports the transition from fragmented global soft law to an integrated, rights-based governance system for food. In this line, Action Track 4 of the global food safety framework provides a detailed plan for incorporating the One Health approach into national food safety systems. risk assessments and surveillance (World Health Organization et al., 2022, pp. 39-43). By strengthening legislative frameworks, scientific capacities and integrated monitoring at the human-animal-plant- environment interface. Action Track 4 supports national states in meeting their constitutional obligations to protect public health and environmental integrity, as well as their international commitments under the WTO SPS Agreement. In this sense, Action Track 4 helps to reduce the legal tensions that have emerged in WTO disputes (such as EC – Hormones, EC – Seal Products analysed in section 4), in which measures intended to protect health, or the environment have been contested as unjustified barriers to trade. Promoting science-based, transparent and proportionate SPS measures reinforces the legitimacy of food safety actions as expressions of constitutional values and fundamental rights while facilitating compliance with international trade law.

4. CASE STUDY: WTO-EU FOOD SAFETY TENSIONS AS HUMAN RIGHTS DILEMMAS

Introductory considerations. Through the SPS Agreement, the WTO plays a central role in global food safety governance. This agreement requires member states to base food safety regulations on scientific risk assessments, while encouraging harmonisation with international standards, such as those set by the Codex Alimentarius Commission. This framework aims to balance states' right to protect public health with their obligation not to create unnecessary barriers to trade (Kastner & Pawsey, 2002). In the context of WTO dispute settlement (Dispute Settlement Understanding, 1995), the tension between trade obligations and human rights protections becomes particularly apparent, as states often encounter legal limitations when attempting to regulate food safety in line with public health, environmental, or ethical considerations³.

The European Union is a leading global standard-setter and a frequent participant in WTO food safety disputes. While the EU's stringent food safety regulations often exceed international norms and have positioned it as a regulatory leader, they have also made it a 'preference outlier'. This limits its influence in global standard-setting and makes it a recurrent party in high-profile WTO disputes, such as EC – Hormones and EC – Seal Products (Young, 2014). These disputes highlight the EU's dual role of advancing high food safety standards domestically and defending them internationally, sometimes in the face of challenges from trading partners and the WTO dispute settlement system. Therefore, while these cases demonstrate functional convergence, they also reveal a normative disjuncture: global trade law incorporates scientific health standards to

_

³ Another interesting line of research would be the analysis of the current crisis of the DSS at the WTO and its plausible reform adopting a right-based orientation in order to constitutionalise the WTO law.

facilitate regulatory harmonisation, yet remains institutionally and legally detached from the human rights regime.

A central normative question emerges: Are these legal disputes fundamentally about the application of rules, or do they reflect deeper power dynamics between regulatory actors? Research suggests that while the SPS Agreement provides a legal framework for resolving disputes, outcomes often hinge on the relative power and preferences of the parties involved, with the EU's regulatory stringency sometimes undermining its ability to shape global norms (Young, 2014). As will be explored in this case study, the analysis debates whether institutions such as the World Trade Organization fulfil 'constitutional functions' by guaranteeing economic freedoms while incorporating non-trade values such as environmental protection and human rights into their legal reasoning and dispute settlement mechanisms (Petersmann, 2008). This evolution has been described as the constitutionalisation of WTO law, and by extension, of international law itself (Trachtman, 2014). From a cosmopolitan One Health perspective, these disputes underscore the need for a more integrated approach that bridges the gap between trade law and human rights, including the right to health and food safety. Current trade law frameworks remain institutionally detached from the human rights regime, leaving significant gaps in the effective protection of food safety at the international level. The EU's experience demonstrates both the potential and the limitations of using trade law to advance broader public health and ethical objectives, highlighting the importance of ongoing debate about the constitutionalisation of WTO law and the incorporation of non-trade values into global food governance (Corini, 2015).

Key Disputes. In the context of global food safety and the right to health, the most instructive and paradigmatic food safety cases are the European Communities -Hormones dispute and the European Communities – Biotech Products dispute. Both cases directly address the intersection of food safety, scientific risk assessment and the limits of national regulatory autonomy under the SPS Agreement (Matsushita, 2009). The EC – Hormones dispute is particularly notable in demonstrating how the WTO strikes a balance between trade liberalisation and the right of states to safeguard public health, and how scientific evidence is considered alongside precautionary measures (Kastner & Pawsey, 2002). The EC – GMOs dispute further highlights the challenges of regulating new food safety technologies and the tensions between domestic risk preferences and international trade obligations (Guzman, 2004). The EC - Seal Products dispute introduces an important ethical dimension by incorporating public morality and animal welfare into the WTO's dispute settlement system, raising questions about the evaluation of non-scientific justifications for trade restrictions within the framework of General Agreement on Tariffs and Trade exceptions⁴. Together, these disputes reveal the limitations of the WTO's current approach, which often treats food safety as a technical trade issue rather than a broader

⁴ Disputes involving food certification and labelling, such as the US – Country of Origin Labelling case and the tuna-dolphin and shrimp-turtle cases, are also relevant for exploring how domestic political pressures and environmental or health concerns shape compliance and regulatory outcomes (Cezar, 2020). In this article, the focus is going to be oriented to disputes where European Union (previous, European Communities) has been sued to narrow geographically the analysis. Nonetheless, there is left here a very appealing line for further research.

human rights or constitutional concern. They also highlight the need for more inclusive, health-centred global governance frameworks.

EC-Hormones (DS26/DS48): Precaution vs. Scientific Risk. The EC-Hormones dispute arose from the European Union's ban on imports of hormone-treated beef, citing concerns over potential public health risks, including long-term effects such as endocrine disruption. The ban, implemented through Council Directive 96/22, covered both synthetic and natural hormones and applied to both domestic and imported meat, with exceptions only for therapeutic and zootechnical purposes (Neugebauer, 2000). The United States and Canada challenged the ban under the SPS Agreement by failing to provide a risk assessment that adequately supported the ban, even though the Appellate Body disagreed with the Panel regarding the consistency of the ban with international standards (Davey, 2006).

The case thus epitomises the tension between the EU's reliance on the precautionary principle, an approach that allows for preventive action in the face of scientific uncertainty, and the WTO's emphasis on science-based risk regulation and procedural consistency among members (Gruszczynski, 2013). While the SPS Agreement does allow for provisional measures in cases of scientific uncertainty (Article 5.7), the EU's ban was not found to meet these requirements (Carlarne, 2007, pp. 306-309).

The article 5.7 of the SPS Agreement is a qualified exemption from the general requirement for science-based measures, designed to prevent abuse of the precautionary principle for protectionist purposes (Ansari and Wartini, 2014). The provision is interpreted narrowly: it applies only when scientific evidence is truly insufficient, and imposes procedural obligations to ensure that provisional measures are temporary and subject to ongoing review (Mercurio and Shao, 2010). The EU's approach was found lacking because it treated scientific uncertainty as absolute, rather than relative and evolving with new evidence (Pauwelyn, 1999).

From a One Health perspective, the EC – Hormones dispute highlights the interconnectedness of human health, animal welfare, and environmental concerns, as the use of growth hormones in livestock production raises systemic questions about food safety, sustainability, and public trust in regulatory institutions. This case exposes the challenges of accommodating higher domestic health standards and public values within the WTO's trade framework, raising important questions about the scope of regulatory autonomy and the legitimacy of international standard- setting processes (Gruszczynski, 2013). A cosmopolitan approach invites reflection on whether international trade rules should provide greater space for domestic regulatory autonomy when it is grounded in higher public health standards and societal values.

EC – Approval and Marketing of Biotech Products (DS291/DS292/DS293). The EC – Biotech Products dispute arose from complaints by the United States, Canada, and Argentina, who challenged the European Union's de facto moratorium on the approval of genetically modified organisms (GMOs). While the EU did not formally prohibit GMOs, it effectively suspended new authorizations and permitted Member States to maintain national bans, citing public health, environmental, and consumer concerns. These

measures, in practice, constituted a prolonged suspension of the EU's approval system, which the complainants argued contravened the WTO's SPS Agreement (Winham, 2009).

In its 2006 report, the WTO Panel found that the EU had breached several SPS provisions, notably by failing to complete risk assessments and by not justifying the moratorium or Member State bans on scientific grounds as required by Article 5.1. The Panel did not rule on the intrinsic safety of GMOs, but its reasoning reinforced the primacy of science-based risk assessment in SPS measures, thereby limiting the EU's reliance on the precautionary principle as a sufficient legal basis for regulatory delay (Bodansky and Lester, 2007).

The Panel considered the concept of 'insufficiency of scientific evidence' under Article 5.7 of the SPS Agreement to be an objective, binary threshold; either there is sufficient evidence to conduct a risk assessment or there is not. However, this approach has been criticised for failing to recognise that scientific sufficiency is inherently relative and context-dependent. Gruszczynski (2008; 2009) argues that sufficiency should be assessed in light of the specific regulatory, cultural and scientific context in which decisions are made and that it may legitimately differ between scientific communities or regulatory authorities in different countries. By contrast, the Panel's method risks imposing a one-size-fits-all standard that does not account for diversity of opinion within the scientific community or the evolving nature of scientific knowledge. In summary, the Panel's absolute and decontextualised approach to scientific sufficiency has been criticised for disregarding the relativity of scientific uncertainty, the legitimacy of culturally sensitive precaution and the evolving nature of scientific knowledge in regulatory decision-making.

The case also illustrates the expanding scope of the SPS Agreement. The panel interpreted it as covering a wide range of health and environmental measures, raising concerns about the WTO's encroachment on domestic regulatory autonomy and environmental protection. The panel's narrow interpretation of the precautionary principle and preference for quantifiable, science-based evidence highlights the ongoing tension between precautionary regulation and the risk-based model embedded in WTO law (Peel, 2006).

From a One Health perspective, the dispute underscores the complex interdependence of human, animal, and environmental health, and the challenge of regulating emerging biotechnologies amid scientific uncertainty and contested risk perceptions (Footer, 2006). The EU's precautionary stance can be understood as an effort to mitigate not only known risks but also emerging and uncertain harms at the intersection of agriculture, food safety, and environmental sustainability, precisely the kind of systemic thinking that One Health encourages.

Finally, the dispute raises normative questions about global regulatory pluralism and the legitimacy of international trade law in constraining democratically endorsed public health and environmental standards. Should international trade law override national or regional regulatory autonomy when it is exercised to protect fundamental values such as health and environmental integrity? The SPS Agreement's strictures may inadequately

accommodate legitimate diversity in public values and risk tolerances, calling for a more flexible, pluralistic approach to global governance that recognizes the legitimacy of precautionary and context-sensitive regulatory choices (Foster, 2008). It calls for a rethinking of global regulatory pluralism, one that accommodates legitimate diversity in public health priorities within a shared framework of human rights and environmental responsibility.

EC – Seal Products (DS400/DS401): Morality, Animal Welfare, and Cultural Rights. The EC – Seal Products dispute is a landmark case that expanded the WTO's engagement with non- economic regulatory objectives, particularly those relating to public morals, animal welfare, and indigenous rights. The EU's ban on seal products was motivated by ethical concerns over animal welfare. Canada and Norway argued that the ban violated non-discrimination obligations under the GATT and the TBT Agreement. This case is notable for its engagement with non-economic regulatory objectives, particularly at the intersection of animal welfare, public morals, and indigenous rights.

Both the Panel and the Appellate Body recognised that animal welfare concerns could be considered part of public morals under Article XX(a) of the GATT. The AB's approach confirmed that the definition of public morals is inherently variable and contextdependent, reflecting the diverse values of WTO members. While the AB upheld the EU's objective, it found that the exceptions in the regime were applied inconsistently, particularly those for indigenous communities (Inuit). This resulted in arbitrary or unjustifiable discrimination, which is contrary to the chapeau of Article XX (Fernández-Pons and Lembo, 2019). The Panel and the Appellate Body clarified that a measure with internal exceptions can still qualify as a legitimate exception under Article XX, provided that the exceptions serve acceptable policy objectives and are not applied in a manner that undermines the measure's core purpose. This marked a significant interpretative development, affirming that WTO members may restrict trade in order to protect ethical values, even when such measures are not strictly science-based or economically motivated. Besides, this new possibility of interpretation highlights the challenge of balancing ethical aims with the fair treatment of affected groups, particularly indigenous peoples whose cultural rights are implicated. The AB's acceptance of pluralistic regulatory values indicates a wider acknowledgement of the validity of ethical and cultural diversity in trade policy. However, it also raises concerns regarding the consistency and predictability of WTO law (Das and Narayan, 2016).

Unlike the EC – Hormones dispute, the Seal Products case centred on non-scientific justifications such as public morals and cultural rights. This challenged the WTO's traditional focus on technical or economic criteria. It affirms the legitimacy of animal welfare and public morals as grounds for trade restrictions, but requires that such measures be applied non-discriminatorily. The case exemplifies the ongoing tension between trade liberalisation, ethical pluralism and the rights of indigenous peoples. It calls for a nuanced approach to global regulatory governance that respects both universal and particularistic values (Fernández-Pons & Lembo, 2019). The One Health approach provides further normative depth by emphasising the connections between animal health, biodiversity conservation and long-term public health – concerns that go beyond

Strict scientific quantification and are aligned with ethical and ecological reasoning. Cosmopolitanism further develops this perspective by posing the question: How should global trade governance accommodate cultural pluralism, collective ethical commitments, and non- economic expressions of public interest? The recognition of moral and cultural grounds as valid justifications for trade restrictions, even if imperfectly applied, suggests that the WTO system is open to rights-based or ethical reasoning to some extent. This reveals a potentially significant development: the possibility of recognising the human rights to health and food security through moral, ecological and cultural imperatives as well as scientific and economic criteria. Such recognition could open the door to a broader normative debate within the WTO about how to reconcile trade rules with the evolving demands of global health and food justice, because the WTO Law only regulates the food safety at the present.

Trade Concerns: EU Pesticides and Sustainable Food. Although no formal WTO dispute has been initiated, several members have expressed ongoing concerns about the EU's restrictive pesticide policies. These include maximum residue limits and hazard-based cut-off criteria under the EU's Green Deal and Farm to Fork Strategy. Critics argue that these measures effectively act as trade barriers, particularly affecting agricultural exports from the Global South.

The European Union's increasingly stringent pesticide regulations, particularly the harmonization and tightening of Maximum Residue Limits (MRLs), have become a focal point of international trade tensions. Non-EU countries such as Brazil, Argentina, and the United States have raised concerns that these measures function as de facto trade barriers, disproportionately affecting agricultural exports from the Global South. Empirical research demonstrates that stricter EU MRLs can significantly reduce bilateral trade flows; for example, a more stringent MRL policy has been shown to decrease US exports of fruits and vegetables to the EU by as much as 13.8% (Hejazi *et al.*, 2022, p. 7). These findings underscore the trade-restricting nature of regulatory divergence in pesticide standards.

At the same time, the EU is defending its pesticide policies, arguing that they are science-based and aligned with the broader sustainability and public health objectives set out in the Green Deal and the Farm to Fork Strategy⁵. In fact, the harmonisation of maximum residue levels across EU Member states has increased intra-EU trade, improved product quality and lowered prices. This suggests that regulatory alignment can have positive outcomes for trade and consumers within the EU (Shingal and Ehrich, 2024)⁶.

⁵ The Farm to Fork Strategy is at the heart of the European Green Deal aiming to make food systems fair, healthy and environmentally friendly. The strategy sets out both regulatory and non-regulatory initiatives, with the common agricultural and fisheries policies as key tools to support a just transition. For r more information, see the European Commission's official page on the *Farm to Fork Strategy*: https://food.ec.europa.eu/horizontal-topics/farm-fork-strategy en (Accessed: 01 July 2025).

⁶ Furthermore, evidence from the EU-Swiss agri-food trade relationship suggests that regulatory homogeneity, whereby trading partners adopt similar MRLs, can enhance food security by increasing product variety, lowering prices and boosting import volumes (Fiankor & Shingal, 2025).

However, the externalization of EU food safety standards through its trade relationships raises complex questions about the global governance of food safety and the normative reach of the SPS Agreement. Discrepancies in pesticide regulations, particularly when developing countries use substances banned in the EU or maintain different MRLs, create significant challenges for exporters and can disrupt international trade. In fact, these regulatory gaps are further complicated by the lack of harmonization with Codex MRLs and the persistence of unauthorized pesticide use in some regions, highlighting the need for greater international cooperation and capacity-building in pesticide management (Kubiak-Hardiman *et al.*, 2022). Looking ahead, the EU's approach to pesticide regulation, which is based on the principles of precaution and sustainability, could provide an opportunity to test the WTO's capacity to support human rights-based and ecologically sustainable food systems. As the EU continues to align its food safety standards with the Sustainable Development Goals and Environmental, Social and Governance frameworks, future legal disputes may challenge the extent to which WTO law can evolve to balance trade facilitation with protecting health, the environment and broader societal values.

Final considerations. What does these disputes show us about the gap between global trade law and global health law? The collaboration between the WTO and the Codex Alimentarius Commission exemplifies both an effort to align global trade law with international food safety norms, and the persistence of structural and normative gaps between trade and health governance. Pursuant to the Agreement on the SPS Agreement, WTO Members are encouraged to base their domestic sanitary and phytosanitary regulations on "international standards, guidelines or recommendations" (Article 3(1)), with the Codex Alimentarius explicitly identified as the relevant standard-setting body for food safety (SPS Agreement, Annex A, para. 3(a)). The Codex, established in the early 1960s by the Food and Agriculture Organization and the World Health Organization, operates as a science-based forum producing voluntary food safety standards with the dual purpose of protecting public health and promoting fair trade practices (Food and Agriculture Organization and World Health Organization, 2020).

However, this *modus of understanding* reveals a significant governance gap: while the Codex standards reflect robust scientific consensus and public health expertise, they lack direct legal force unless the Codex is integrated into WTO adjudication. Within the WTO framework, the Codex standards serve primarily as reference points for determining whether national measures are 'necessary' or 'justified' under SPS Agreement. In disputes such as EC - Hormones (DS26, DS48) and EC - Sardines (DS231), the Dispute Settlement Body has relied on Codex standards to assess whether domestic food safety regulations constitute disguised trade barriers (World Trade Organization Appellate Body, 1998, WT/DS26/AB/R and WT/DS48/AB/R; WTO Appellate Body, 2002, WT/DS231/AB/R). This use, however, subordinates public health protection to the trade logic of non-discrimination and market access, rather than recognizing food safety as an enforceable component of the human right to health (Hunt and Backman, 2008).

Furthermore, the Codex Alimentarius operates as a soft law instrument. Therefore, the Codes runs without mechanisms for enforcement, participatory oversight, or individual redress. Its standards are not binding under international human rights law and are not

framed within a rights-based paradigm. The SPS Agreement itself conceptualizes food safety primarily in terms of technical barriers to trade, rather than through a framework of public health accountability or state obligations to protect the right to safe food (SPS Agreement, Article 2 and Annex A, para. 1).

As illustrated in the preceding case study, this tension is central to a broader scholarly debate, this tension is central to a broader scholarly debate over whether institutions such as the WTO can be understood to perform 'constitutional functions' - namely, by guaranteeing economic freedoms while also incorporating non-trade values such as environmental protection and human rights into their legal reasoning and dispute settlement mechanisms (Petersmann, 2008). This process has been described as the 'constitutionalisation' of WTO law and, by extension, of international law itself (Trachtman, 2014). The extent to which food safety is conceptualized as a trade measure rather than a human right thus raises important questions about the legitimacy, limits, and evolving role of global trade institutions in protecting non-economic values.

5. FOOD SAFETY AND ESG: TOWARD A RIGHTS-BASED FRAMEWORK

Preliminary considerations. The intersection between food safety and human rights is often overlooked in global governance debates, where the focus tends to gravitate toward food security and access. Yet food safety, defined as the assurance that food will not cause harm to the consumer when prepared and consumed as intended, is an equally fundamental dimension of the right to adequate food. While food security concerns availability, access, and stability of food supply; food safety concerns the quality and safety of what is consumed. This distinction is essential: a person may have regular access to food and still suffer serious violations of their right to health due to contaminated or unsafe food products. The Office of the High Commissioner for Human Rights has affirmed that the right to food encompasses not only the availability of food but also its safety and acceptability, as reflected in General Comment No. 12 of the Committee on Economic, Social and Cultural Rights, which interprets Article 11 of the International Covenant on Economic, Social and Cultural Rights to include food safety as a core component of the right to adequate food (Ayala and Meier, 2017). The Universal Declaration of Human Rights, Article 25, also recognizes the right to a standard of living adequate for health and well-being, including food, which implicitly requires that food be safe and nutritious (United Nations General Assembly, 1948).

From a governance perspective, the Codex Alimentarius, developed by the FAO and WHO, has become the central reference point for international food safety standards and is explicitly recognized in the WTO's SPS Agreement. However, the voluntary and technocratic nature of these instruments raises questions about their normative robustness and ability to advance a rights- based approach, as they often lack direct enforceability and may not fully integrate human rights obligations (Milian Gómez, 2024). In this context, ESG frameworks offer an emerging avenue for embedding food safety obligations within broader accountability systems, but their contribution remains contested and incomplete, as they are typically non-binding and rely on market and reputational incentives rather than legal compulsion (Pettoello-Mantovani and Olivieri, 2022).

The EU's approach to pesticide regulation, grounded in principles of precaution and sustainability, may offer an important testing ground for assessing whether the WTO can accommodate food systems that prioritize human rights and ecological values. As the EU continues to align its food safety standards with the SDGs and ESG frameworks, future legal disputes may challenge the current balance between trade liberalization and the protection of health, the environment, and other public goods (Pettoello-Mantovani and Olivieri, 2022). These emerging tensions invite a shift in how food safety is conceptualized: not simply as a technical trade issue, but as an integral element of the human right to health and dignity, as recognized in evolving international doctrine and legal scholarship (Milian Gómez, 2024).

This shift also demands broader governance reflection. How can international legal frameworks ensure that food is not only available and accessible, but also safe and health-promoting? Can the fragmentation between trade law, public health, and human rights be overcome through integrated tools such as ESG reporting and SDG-based governance mechanisms? The literature suggests that a rights-based approach—grounded in both legal and institutional innovation—can embed food safety into the evolving architecture of global sustainability and accountability, but this requires explicit recognition of food safety as a legal right and the harmonization of international, regional, and national legal frameworks (Milian Gómez, 2024).

ESG frameworks. Although ESG frameworks originated in the fields of corporate governance and financial risk management, they have increasingly influenced global discussions on sustainability in food systems, particularly with regard to corporate responsibility for public health and food safety. The growing importance of ESG in the food sector is evident in recent EU regulatory proposals and sustainability action plans, which emphasise the necessity for companies to incorporate environmental, social and governance considerations into their operations and reporting practices (Gossain, 2023)7. Similarly, the SDGs provide a shared normative agenda that promotes longterm ecological balance, cross-sectoral coordination and the protection of fundamental rights, including the right to health. Together, both the ESG and the SDGs create a new space in which food safety can be incorporated into institutional accountability and sustainability metrics. Increasing pressure from consumers, investors and regulators for transparent ESG reporting is prompting food companies to adopt robust sustainability and food safety practices (Gołębiewski, 2023). Nonetheless, the complexity and diversity of ESG regulations across jurisdictions present significant legal and ethical challenges, highlighting the need for clearer, harmonised guidelines to ensure effective, accountable implementation (Gossain, 2023).

The ESG and Codex Alimentarius. Although ESG standards are often described as soft law due to their voluntary nature and the risk of superficial compliance, such as

_

⁷ These frameworks are reinforced by international initiatives such as the Task Force on Climate-Related Financial Disclosures and the Paris Agreement, which have accelerated the harmonisation of reporting standards and provided stronger legal foundations for ESG compliance worldwide. Ia One Health Approach to Food safety and the Right to

greenwashing, they are becoming increasingly influential in shaping corporate behaviour. This influence stems from investment screening, reputational incentives, and pressure from consumers and civil society. For example, the UN Principles for Responsible Investment encourage investors to consider ESG risks, including those related to food safety and public health, in their decision-making processes. This, even being soft law, creates a form of market-driven accountability. In this context, internationally recognised food safety standards, such as those of the Codex Alimentarius and the WTO's SPS Agreement, can be incorporated into corporate supply chain policies, quality assurance protocols, and ESG reporting (Gregory *et al.*, 2024).

Although these ESG frameworks and the Codex Alimentarius are not legally binding, firstly, the ESG frameworks may thus serve as a complementary governance mechanism even in the absence of formal legal obligation, providing a pathway for food safety to become part of private sector compliance and transparency. Secondly, the Codex Alimentarius offers a comprehensive set of principles for good hygiene practices, which can guide food producers in aligning their processes with both food safety and ESG objectives. However, the diversity and lack of harmonization in ESG ratings and reporting methodologies present ongoing challenges, underscoring the need for clearer guidelines and more robust integration of Codex principles into ESG frameworks to ensure meaningful and accountable implementation (Gregory *et al.*, 2024). In this context, the SPS Agreement and the jurisprudence of the WTO in its implementation can serve as a guide to move from soft law to more binding law.

Soft Law as a Pathway to Hard Law: The Normative Evolution of ESG Frameworks. Despite critiques that ESG standards lack binding authority and are vulnerable to greenwashing, it is essential to recognize that soft law has historically served as a stepping stone to binding obligations in international law. For example, in international environmental law, the progression from the non-binding Stockholm Declaration (1972) to the Rio Declaration (1992), and ultimately to legally binding agreements like the Paris Agreement (2015), illustrates how soft law can lay the groundwork for subsequent hard law instruments. Similarly, in the domain of business and human rights, the UN Guiding Principles on Business and Human Rights (2011), though non- binding, have significantly influenced national legislation, corporate policies, and due diligence obligations, as seen in the development of the EU's Corporate Sustainability Due Diligence Directive (European Union, 2024).

Soft law instruments, such as ESG frameworks, offer flexibility and broad participation, allowing ambitious norms to be established and political pressure to be exerted on laggards, which can accelerate the transition to binding regulation. In the context of global health and sustainability, soft law policies have shaped global norms and expectations, but, especially in crisis situations, their limitations have prompted calls for more robust, enforceable legal frameworks. Thus, ESG frameworks can be seen as incubators of normative consensus. When reinforced by political will, civil society engagement, and investor expectations, they may evolve into binding regulatory regimes at domestic, regional, or international levels. From a food safety perspective, this dynamic offers a promising avenue for embedding Codex Alimentarius principles, sustainability

goals, and the right to health into more robust legal and policy frameworks, leveraging the complementary strengths of both soft and hard law to advance global standards and accountability (Meier *et al.*, 2025).

ESG and Corporate Behaviour: Internalizing Food Safety through Market **Accountability.** ESG frameworks have reshaped corporate accountability by pushing companies to uphold food safety standards, not just financial performance. In the food sector, pressure from investors, consumers, and reputational risks increasingly incentivises alignment with Codex Alimentarius guidelines. Many global food companies have already begun to voluntarily implement Codex- aligned standards. For example: adopting Hazard Analysis and Critical Control Points systems, carrying out food safety audits based on Codex guidelines and disclosing risk management practices (Camilleri, 2025). Empirical research shows that continuous environmental, social and governance management activities in the food sector enhance corporate reputation, consumer trust and intention to buy again, and demonstrates that market-driven accountability can be a powerful motivator for food safety compliance (Tian et al., 2024). Furthermore, ESG disclosures are increasingly being used to communicate responsible food safety practices to stakeholders. These disclosures are guided by standards such as those set out by the Global Reporting Initiative and the Sustainability Accounting Standards Board (Camilleri, 2025). This process of internalization demonstrates how soft norms, when embedded in ESG metrics and investor expectations, can influence corporate behaviour. Although it cannot replace formal legal frameworks, it offers an alternative mode of governance, particularly in a globalised food system where corporations frequently operate beyond the jurisdiction of national regulators (Tian et al., 2024).

Final Considerations. In order to strengthen the nascent alignment between global food safety standards and corporate responsibility, public institutions and international organisations should actively support the development of ESG reporting frameworks that explicitly incorporate the principles of the Codex Alimentarius. By encouraging firms to embed food safety into their risk assessments, sustainability metrics, and governance structures, such guidance can promote a more coherent and accountable model of food system governance. This effort would not only enhance transparency and consumer trust, but also contribute to the recognition of food safety as a core element of the human right to health. In doing so, it could help bridge the normative divide between trade law, public health, and human rights-advancing a cosmopolitan vision of global governance in which private and public actors share responsibility for upholding health and sustainability standards across borders (Camilleri, 2025).

6. Conclusions

Reframing Global Food Safety through the Right to Health. This article explores the normative, legal and institutional disconnect between food safety regulation and the international human rights framework, with a particular focus on the right to health. Although food safety is an essential aspect of the right to adequate food, it is marginalised within the global legal framework, primarily due to the ongoing fragmentation of health governance based on sovereignty. In this context, the absence of binding global food safety

standards is not merely a technical oversight, but reflects deeper structural asymmetries in the global order, such as those between trade and health law, soft law and enforceable norms, and market efficiency and ethical responsibility.

The right to health, as set out in international human rights law (particularly Article 12 of the International Covenant on Economic, Social and Cultural Rights), obliges states to ensure access to safe and nutritious food. However, there is currently no binding international legal instrument that integrates food safety as an enforceable legal entitlement beyond national borders. As demonstrated, the fragmented normative architecture, characterised by the WTO SPS Agreement, the Codex Alimentarius and ESG soft law, fails to coalesce around a unified, rights-based framework for food governance. This regulatory gap suggests the need for a paradigm shift, from sectoral governance based on trade facilitation, to an integrated regime that embeds health and safety within global public goods frameworks. The One Health paradigm offers a compelling means of achieving this transformation, particularly when reinforced by cosmopolitan legal theory and ESG accountability mechanisms.

From Soft Law Aspirations to Integrated Global Governance: The Role of **One Health.** The One Health approach provides a comprehensive understanding of the interconnected nature of human, animal, and environmental health. In practice, however, it remains grounded in soft law instruments, declarations and institutional coordination platforms, lacking enforceable mechanisms. This limits its capacity to address cross-border threats with binding force. While the One Health Joint Plan of Action (2022–2026) and the Global One Health Index represent significant progress, they still operate within voluntary regimes. Can One Health evolve from a soft law discourse into a normative foundation for integrated governance? This article argues that it can, but only if it is anchored within binding legal obligations, particularly the right to health. The legal codification of One Health principles would strengthen the accountability of both states and corporations and provide a coherent framework for addressing emerging risks, such as outbreaks of zoonotic diseases, antimicrobial resistance and collapse of the food system. One Health is uniquely positioned to bridge current governance silos. Its integrative ethos, based on scientific cooperation and cross-sectoral coordination, aligns with the pluralistic and ethical principles of cosmopolitan legal theory. Integrating One Health into international legal frameworks through SDG alignment, ESG integration and reinterpreting existing treaties offers a pragmatic way to enhance legal coherence, improve public health resilience and safeguard shared ecological resources.

Key findings from the WTO-EU food safety case study: Are these disputes about rules or about power? An analysis of WTO-EU food safety disputes (EC-Hormones, EC-Biotech, EC- Seal Products and pesticide trade concerns) reveals an ongoing tension: is the WTO legal order equipped to protect health as a human right, or is it predisposed to prioritise trade liberalisation? While WTO jurisprudence has slowly evolved to recognise certain non-economic values (e.g. public morals and animal welfare), the evidence suggests that its core institutional design continues to favour market access and scientific risk rationality over precautionary, ethical, or context-sensitive approaches to regulation.

This asymmetry is particularly apparent in disputes where the EU seeks to enforce higher regulatory standards based on precautionary, cultural or ethical grounds. While some of these standards reflect cosmopolitan ideals, such as transnational environmental responsibility or animal welfare, others are considered neo-colonial or protectionist by affected trading partners, particularly those in low- and middle-income countries. From a cosmopolitan perspective, these tensions reveal not also legal inconsistencies; they expose the global power dynamics inherent in rule-based multilateralism. The outcomes of disputes most of the time depend on relative bargaining power and institutional asymmetries, rather than the merits of the public health claims presented.

The EC – Seal Products case is particularly illustrative in this light because while the WTO accepted public morals as a valid basis for trade restrictions, it also demonstrated how difficult it is to apply pluralistic values within a system that demands constant jurisprudence, non- discrimination and scientifically based justification. Meanwhile, the other two disputes analysed in this paper, Biotech and Hormones, revealed the constraints imposed by the narrow construction of scientific sufficiency in the SPS Agreement, which fails to account for evolving or contested scientific knowledge. Ultimately, these disputes demonstrate that WTO law does not adequately accommodate the diversity of legitimate food safety standards, precautionary approaches and public health priorities across jurisdictions. A cosmopolitan trade regime would require respect for local contexts and vulnerabilities, including culturally embedded food preferences and ethical norms, recognition of shared transnational responsibilities – especially with regard to ecological protection and public health risks that transcend borders – and a regulatory approach that balances science, precaution and ethics, rather than defaulting to a uniform model of scientific rationality.

Normative implications and policy reform directions. This article contributes to the growing body of scholarship calling for a rights-based reconfiguration of global food governance. Several normative implications and policy directions emerge: First, food safety should be reconceptualised as a human right, rather than merely a trade concern. This would require revising the WTO SPS Agreement to include references to the right to health, and ensuring that trade rules align with international human rights law. Second, legally integrate One Health into global health and food governance. This could be achieved through new treaties, interpretative declarations or amendments to existing instruments (e.g. the WHO International Health Regulations) to include food safety and the environmental determinants of health.

The ESG agenda should move towards establishing binding obligations. As the article has shown, ESG frameworks currently operate as market-driven accountability mechanisms. However, they have significant normative potential. By incorporating Codex Alimentarius standards, One Health principles, and human rights metrics into ESG reporting, governments and international organisations can promote compliance within the private sector and achieve public objectives. There should be institutional convergence between trade and health governance bodies, including stronger collaboration between the WTO, the Codex, the WHO and human rights institutions. Joint interpretative guidance, cross-referencing standards, and inclusive participation by health and civil society actors

in trade forums would contribute to legal coherence. Global democratic legitimacy can be enhanced through participatory governance. The legitimacy crisis in international law stems from fragmented norms and the exclusion of affected communities from decision-making processes. Cosmopolitan governance requires legal integration, democratic accountability and inclusion.

Towards a Cosmopolitan Vision of Global Food Governance. This article concludes that food safety should be better regulated and that the normative basis of global governance must shift to prioritise health protection – both individual and ecological. This requires us to recognise food safety as a shared human right, embedded in a framework of collective responsibility and legal accountability that transcends borders. Cosmopolitanism provides the ethical and legal framework for this change. It encourages us to move beyond the limitations of state-based governance and articulate a framework in which everyone, regardless of geography or citizenship, has access to safe, nutritious, and ethically produced food. It also requires states and corporations to recognise the interdependencies of human, animal and environmental health, and to accept that the costs of inaction will be shared by all. The road ahead is complex. It involves legal reform, institutional redesign and cultural transformation. However, the convergence of health, environmental, and trade challenges presents a unique opportunity to transition from fragmented, reactive governance to an integrated, preventive, and ethically grounded model. This model would protect health not just as a policy aspiration, but as a fundamental global right.

FUNDING

This research is conducted within the framework of the Research Group "Challenges and Opportunities of ESG Criteria in the Business Environment" (ESGEMP), as part of the Interuniversity Programme on the Culture of Lawfulness (4Trust-CM, Reference: PHS-2024/PHHUM-65), funded by the Community of Madrid; and within the framework of the research project "Towards a New Governance in the EU: From CSR to ESG Factors' (PP2024_18), funded under the 2024 Call for Internal Research Projects of the Universidad Pontificia Comillas".

REFERENCES

Doctrinal References & Official Documents

- ANSARI, A.H. AND WARTINI, S. (2014). "Precautionary principle under the SPS Agreement: A critical exposition". *Journal of International Trade Law and Policy*, 13(1),pp. 19-43. Available at: https://www.researchgate.net/publication/282720825 (Accessed: 11 July 2025).
- ARCURI, A. (2015). Global food safety standards. The evolving regulatory epistemology at the intersection of the SPS Agreement and the Codex Alimentarius Commission. In P. DELIMATSIS (ed.), *The Law, Economics and Politics of International Standardisation. Cambridge International Trade and Economic Law.* Cambridge University Press, pp. 79-103.

- ARRIOLA ECHANIZ, N. (2008). "El cosmopolitismo como teoría defensora del gobierno mundial". Estudios de Deusto: Revista de Derecho Público, 56(2), pp. 11-72
- ARRIOLA ECHANIZ, N. (2017). "Las consecuencias de la globalización en el derecho constitucional: aportaciones de la miríada de corrientes doctrinales". Revista de Estudios Políticos, 178, pp. 171-188.
- ARRIOLA ECHANIZ, N. (2019). "¿Más allá del Estado nacional? Una revisión de la doctrina cosmopolita". Revista de Estudios Políticos, 183, pp. 243-259.
- AYALA, A. AND MEIER, B. M. (2017). "A human rights approach to the health implications of food and nutrition insecurity". Public Health Reviews, 38. Available at: https://doi.org/10.1186/s40985-017-0056-5 (Accessed: 30 June 2025).
- BECK, U. (2002). Libertad o capitalismo: conversaciones con Johannes Willms. Barcelona: Paidós.
- BECK, U. (2005). La mirada cosmopolita o la guerra es la paz. Barcelona: Paidós.
- BODANSKY, D. AND LESTER, S. (2007). "European Communities Measures affecting the approval and marketing of biotech products. WT/DS291/R, WT/DS292/R, & WT/DS293/R". *American Journal of International Law*, 101, pp. 453-459. Available at: https://doi.org/10.1017/S0002930000030177 (Accessed: 30 June 2025).
- CABRERA, L. (2020). The humble cosmopolitan: rights, diversity, and trans-state democracy. Oxford: Oxford University Press.
- CAMILLERI, M. A. (2025). "Sustainability accounting and disclosures of responsible restaurant practices in environmental, social and governance (ESG) reports". *International Journal of Hospitality Management*, *126*, e104051. Available at: https://doi.org/10.1016/j.ijhm.2024.104051 (Accessed: 09 June 2025).
- CARLARNE, C. P. (2007). "From the USA with love: sharing home-grown hormones, GMOs and clones with a reluctant Europe". *Environmental Law*, *37*(2), pp. 301-337. Available at: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=1678608 (Accessed: 10 July 2025).
- CEZAR, R. (2020). "Food certification, domestic politics and international trade: the US compliance response in three WTO disputes". *Environmental Politics*, *29*, pp. 317-335. Available at: https://doi.org/10.1080/09644016.2019.1565462 (Accessed: 07 July 2025).
- CORINI, A. (2015). Human right to food: some reflections. In BALDINI, C., MARAN, R. and VITTUARI, M. (eds.), *Envisioning a future without food waste and food poverty*. Leiden: Wageningen Academic Publishers, pp. 317-322. Available at: https://doi.org/10.3920/978-90-8686-820-9 39 (Accessed: 10 July 2025).
- DAHRENDORF, R. (2005). En busca de un nuevo orden: una política de la libertad para el siglo XXI. Barcelona: Paidós Ibérica.

- DAS, A. S. AND NARAYAN, A. V. (2016). "Settling the debate of animal welfare, public morals and trade: in the light of the *EC-Seal Products* case". *Global Trade and Customs Journal*, 11(6), 267-279. Available at: https://doi.org/10.54648/gtcj2016036 (Accessed: 30 June 2025).
- DAVEY, W. J. (2006). Reflections on the Appellate Body decision in the Hormones case and the meaning of the SPS Agreement. In BERMANN, G. A. and MAVROIDIS, P. C. (eds.), *Trade and human health and safety*. Cambridge: Cambridge University Press, pp. 118-132.
- DE SCHUTTER, O. (2014). *The Transformative Potential of the Right to Food*. Report of the Special Rapporteur on the Right to Food, A/HRC/25/57. United Nations Human Rights Council.
- DE SOUSA SANTOS, B. (2005). El milenio huérfano: ensayos para una nueva cultura política. Madrid: Trotta.
- DE SOUSA SANTOS, B. AND RODRÍGUEZ GARAVITO, C. A. (2007). *El derecho y la globalización desde abajo: hacia una legalidad cosmopolita*. Barcelona: Anthropos.
- DUDLEY, W. AND ENGELHARD, K. (eds.) (2011). *Immanuel Kant: key concepts*. 1st edn. Durham: Acumen Publishing.
- FALK, R. (2002). *La globalización depredadora: una crítica*. Madrid: Siglo XXI de España.
- FERNÁNDEZ-PONS, X. AND LEMBO, C. (2019). The case *EC Seal Products*: the WTO Dispute Settlement System before a "trilemma" between free trade, animal welfare, and rights of Indigenous Peoples. In DO AMARAL JÚNIOR, A., DE OLIVEIRA SÁ PIRES, L. M. and LUCENA CARNEIRO, C. (eds.), *The WTO Dispute Settlement Mechanism*. Cham: Springer, pp. 305-319. Available at: https://doi.org/10.1007/978-3-030-03263-0_20 (Accessed: 11 July 2025).
- FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS AND WORLD HEALTH ORGANIZATION (2020). *Understanding the Codex Alimentarius* 4th edn). Rome: FAO and WHO. Available at: https://www.fao.org/fsnforum/resources/trainings-tools-and-databases/understanding-codex-alimentarius (Accessed: 30 June 2025).
- FIANKOR, D. D. AND SHINGAL, A. (2025). "Pesticide regulatory homogeneity and firms' import decisions: evidence from EU-Swiss agri-food trade". *Journal of Agricultural Economics*, 76, pp. 338-358. Available at: https://doi.org/10.1111/1477-9552.12623 (Accessed: 11 July 2025).
- FIDLER, D. P. (2004). "Germs, governance, and global public health in the wake of SARS". *The Journal of Clinical Investigation*, 113(6), pp. 799-804.
- FOOTER, M. E. (2006). Post-normal science in the multilateral trading system: social science expertise and the EC Biotech Panel. Hebrew University International

- Law Research Paper No. 15-06. Available at: https://ssrn.com/abstract=953235 (Accessed: 30 June 2025).
- FORTIN, N. D. (2023). Global governance of food safety: the role of the FAO, WHO, and Codex Alimentarius in regulatory harmonization. In M. T. ROBERTS (ed.), Research Handbook on International Food Law. Edward Elgar Publishing, pp. 227-242.
- FOSTER, C. E. (2008). "Public opinion and the interpretation of the World Trade Organisation's Agreement on Sanitary and Phytosanitary Measures". *Journal of International Economic Law*, *11*(2), pp. 427-458. Available at: https://doi.org/10.1093/jiel/jgn011 (Accessed: 11 July 2025).
- GOŁĘBIEWSKI, J. (2023). "The significance and essence of ESG in the food sector in Poland". *Annals of the Polish Association of Agricultural and Agribusiness Economists*, 25(1), pp. 75-90. Available at: https://doi.org/10.5604/01.3001.0016.2825 (Accessed: 30 June 2025).
- GOSSAIN, R. (2023). "Legal underpinnings drive ESG amidst global flux". *Paradigm*, 27(1), pp. 27-46. Available at: https://ideas.repec.org/a/sae/padigm/v27y2023i1p27-46.html (Accessed: 16 June 2025).
- GREGORY, G., LERMEN, F. H. AND ECHEVESTE, M. E. S. (2024). "Toward food safety-driven process design: a systematic review and research agenda". *Critical Reviews in Food Science and Nutrition*, 1-14. Available at: https://doi.org/10.1080/10408398.2024.2400590 (Accessed: 12 July 2025).
- GRUSZCZYNSKI, L. (2008). SPS measures adopted in case of insufficiency of scientific evidence where do we stand after *EC Biotech Products* case? In CHAISSE, J. and BALMELLI, T. (eds.), *Essays on the future of the World Trade Organization, vol. 2: The WTO judicial system contributions and challenges*. Lugano/Brussels: Edis, pp. 91-140. Available at: https://ssrn.com/abstract=1100061 (Accessed: 16 June 2025).
- GRUSZCZYNSKI, L. (2009). "Insufficiency of scientific evidence under Article 5.7 of the SPS Agreement: some remarks on the panel report in the *EC Biotech Products* case". *University of Ottawa Law & Technology Journal*, 6(1-2), pp. 53-78. Available at: https://ssrn.com/abstract=2034565 (Accessed: 16 June 2025).
- GRUSZCZYNSKI, L. (2013). Standard of review of health and environmental regulations by WTO panels. In VAN CALSTER, G. and PRÉVOST, D. (eds.), *Research handbook on environment, health and the WTO*. Cheltenham: Edward Elgar Publishing, pp. 731-758. Available at: https://doi.org/10.4337/9781781006146.00032 (Accessed: 11 July 2025).
- GUZMAN, A. (2004). "Food fears: health and safety at the WTO". Virginia Journal of International Law, 45, pp. 1-40.
- HEJAZI, M., GRANT, J. H. AND PETERSON, E. (2022). "Trade impact of maximum residue limits in fresh fruits and vegetables". *Food Policy*, *106*, e102203. Available at: https://doi.org/10.1016/j.foodpol.2021.102203 (Accessed: 30 June 2025).

- HELD, D. (2010). Cosmopolitanism: ideals and realities. Cambridge: Polity Press.
- HELD, D. (2016). "Elements of a theory of global governance". *Philosophy and Social Criticism*, 42(9), pp. 837–846. Available at: https://doi.org/10.1177/0191453716659520 (Accessed: 30 June 2025).
- HUNT, P. (2016). "Interpreting the international right to health in a human rights-based approach to health". *Health and Human Rights*, *18*, pp. 109-130.
- HUNT, P. AND BACKMAN, G. (2008). "Health systems and the right to the highest attainable standard of health". *Health and Human Rights*, *10*(1), pp. 81-92. Available at: https://www.hhrjournal.org/2013/09/13/health-systems-and-the-right-to-the-highest-attainable-standard-of-health/ (Accessed: 30 June 2025).
- INTERNATIONAL FINANCE CORPORATION (2004). Who cares wins: connecting financial markets to a changing world. IFC Advisory Services in Environmental and Social Sustainability issue brief. Washington, D. C.: World Bank Group. Available at: http://documents.worldbank.org/curated/en/444801491483640669 (Accessed: 01 July 2025).
- JASANOFF, S. (2004). *States of Knowledge: The Co-Production of Science and the Social Order*. London: Routledge.
- KASTNER, J. AND PAWSEY, R. (2002). "Harmonising sanitary measures and resolving trade disputes through the WTO-SPS framework. Part I: a case study of the US-EU hormone-treated beef dispute". *Food Control*, *13*(1), pp. 49-55. Available at: https://doi.org/10.1016/S0956-7135(01)00023-8 (Accessed: 07 July 2025).
- KELSEN, H. (2001). *Introducción a la teoría pura del derecho*. 3rd edn, rev. y corr. México: Universidad Nacional Autónoma de México. With contributions by García Belaunde, D., Rabasa, E. O. and Palomino Manchego, J. F.
- KUBIAK-HARDIMAN, P., HAUGHEY, S., MENEELY, J., MILLER, S., BANERJEE, K. AND ELLIOTT, C. (2022). "Identifying gaps and challenges in global pesticide legislation that impact the protection of consumer health: rice as a case study". *Exposure and Health*, *15*, pp. 597-618. Available at: https://doi.org/10.1007/s12403-022-00508-x (Accessed: 30 June 2025).
- LANGE, D. (2021). "The relation between EU food safety policy, Codex Alimentarius and WTO. Evolution and current challenges". *Ernahrungs Umschau*, 68(10), pp. 204-209.
- MATSUSHITA, M. (2009). WTO dispute cases relating to food safety issues. In VAN DEN BOSSCHE, P. and ZDOUC, W. (eds.), *The WTO and international trade law/dispute settlement*. Bingley: Emerald Group Publishing, vol. 6, pp. 283-302. Available at: https://doi.org/10.1108/S1574-8715(2009)0000006014 (Accessed: 07 July 2025).
- MEIER, B. M., FINCH, A. AND HABIBI, R. (2025). "Global health law: between hard and soft law". *Journal of Law, Medicine & Ethics*, *53*(S1), pp. 11-15. Available at: https://doi.org/10.1017/jme.2025.21 (Accessed: 09 June 2025).

- MERCURIO, B. AND SHAO, D. (2010). "A precautionary approach to decision making: the evolving jurisprudence on Article 5.7 of the SPS Agreement". *Trade, Law and Development*, 2(2), 195-222. Available at: https://www.researchgate.net/publication/228122217 (Accessed: 11 July 2025).
- MILIAN GÓMEZ, J. F. (2024). "Rethinking the human right to food from a single perspective to a four-fold legal interpretation". *Journal of Human Rights Practice*, *16*(2), pp. 589-602. Available at: https://doi.org/10.1093/jhuman/huad047 (Accessed: 30 June 2025).
- MOKA-MUBELO, W. (2017). A cosmopolitan human rights regime'. In CORRADETTI, C. (ed.), *Reconciling law and morality in human rights discourse*. Philosophy and Politics Critical Explorations, vol. 3. Cham: Springer, pp. 169-199. Available at: https://doi.org/10.1007/978-3-319-49496-8 7 (Accessed: 08 July 2025).
- NEUGEBAUER, R. (2000). "Fine-tuning WTO jurisprudence and the SPS Agreement: lessons from the beef hormone case". *Law and Policy in International Business*, 31(4), 1255-1284.
- ONE HEALTH HIGH-LEVEL EXPERT PANEL (OHHLEP), ADISASMITO, W. B., ALMUHAIRI, S., BEHRAVESH, C. B., BILIVOGUI, P., BUKACHI, S. A. *et al.* (2022). "One Health: A new definition for a sustainable and healthy future". *PLoS Pathogens*, *18*(6), e1010537. Available at: https://doi.org/10.1371/journal. ppat.1010537 (Accessed: 03 July 2025).
- PAUWELYN, J. (1999). "The WTO Agreement on Sanitary and Phytosanitary (SPS) Measures as applied in the first three SPS disputes: EC-Hormones, Australia-Salmon and Japan-Varietals". *Journal of International Economic Law*, *2*(4), pp. 641-664. Available at: https://doi.org/10.1093/jiel/2.4.641 (Accessed: 07 July 2025).
- PEEL, J. (2006). "A GMO by any other name... might be an SPS risk! Implications of expanding the scope of the WTO Sanitary and Phytosanitary Measures Agreement". *European Journal of International Law*, 17(5), 1009-1031. Available at: https://doi.org/10.1093/ejil/chl033 (Accessed: 11 July 2025).
- PETERS, A. (2018). Global constitutionalism: the social dimension. In SUAMI, T., KUMM, M., PETERS, A. and VANOVERBEKE, D. (eds.), *Global constitutionalism from European and East Asian perspectives*. Cambridge: Cambridge University Press, pp. 277–350.
- PETERSMANN, E. U. (2008). "Human rights, international economic law and constitutional justice: a rejoinder" *European Journal of International Law*, 19(5), pp. 955–960.
- PETERSMANN, E. U. (2012). International economic law in the 21st century: constitutional pluralism and multilevel governance of interdependent public goods. Oxford: Hart Publishing.
- PETERSMANN, E. U. (2013). Human rights and the law of the World Trade Organization. In COTTIER, T. and PESTALOZZI, M. (eds.), *Linking trade and human rights:* new policy space in hard economic times. Cheltenham: Edward Elgar, pp. 29-56.

- PETERSMANN, E. U. (2017). Cosmopolitan constitutionalism: linking local engagement with international economic law and human rights. In BIUKOVIC, L. and POTTER, P. B. (eds.), *Local engagement with international economic law and human rights*. Cheltenham: Edward Elgar Publishing, pp. 26-54. Available at: https://doi.org/10.4337/9781785367199.00008 (Accessed: 08 July 2025).
- PETTOELLO-MANTOVANI, C. AND OLIVIERI, B. (2022). "Food safety and public health within the frame of the EU legislation". *Global Pediatrics*, 2, e100020. Available at: https://doi.org/10.1016/j.gpeds.2022.100020 (Accessed: 12 July 2025).
- ROZPEDOWSKI, J. (2019). Between sovereign prudence and global jurisprudence: the evolution of supranational courts and cosmopolitan norms in human rights discourse and state practice. Doctoral thesis, Durham University. Available at: https://etheses.dur.ac.uk/13099/ (Accessed: 08 July 2025).
- SHINGAL, A. AND EHRICH, M. (2024). "The EU's pesticides MRLs harmonization: effect on trade, prices and quality". *Food Policy*, *125*, e102634. Available at: https://doi.org/10.1016/j.foodpol.2024.102634 (Accessed: 11 July 2025).
- TIAN, H., YOON, Y. AND JUNG, J. (2024). "The impact of ESG management activities of food service companies on corporate image, corporate reputation and repurchase intention: focusing on Luckin Coffee in China". *The Academic Society of Global Business Administration*, 21(6), 285-307. Available at: https://doi.org/10.38115/asgba.2024.21.6.285 (Accessed: 30 June 2025).
- TRACHTMAN, J. P. (2014). The economic structure of the law of international organizations". *Chicago Journal of International Law*, 15(1), pp. 162-194. Available at: https://chicagounbound.uchicago.edu/cjil/vol15/iss1/9 (Accessed: 30 June 2025).
- WINHAM, G. R. (2009). "The GMO Panel: applications of WTO law to trade in agricultural biotech products". *Journal of European Integration*, *31*(3), pp. 409-429. Available at: https://doi.org/10.1080/07036330902782261 (Accessed: 11 July 2025).
- WORLD HEALTH ORGANIZATION, FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS, UNITED NATIONS ENVIRONMENT PROGRAMME AND WORLD ORGANISATION FOR ANIMAL HEALTH (2022). One Health Joint Plan of Action (2022-2026): Working together for the health of humans, animals, plants and the environment. Geneva: World Health Organization. Available at: https://www.who.int/publications/i/item/9789240059139 (Accessed: 03 July 2025).
- YOUNG, A. R. (2014). "Europe as a global regulator? The limits of EU influence in international food safety standards". *Journal of European Public Policy*, *21*(6), pp. 904-922. Available at: https://doi.org/10.1080/13501763.2014.910871 (Accessed: 10 July 2025).
- ZHANG, Q., LIU, J., HAN, L., LI, X., ZHANG, C., GUO, Z., CHAO, A., WANG, C., WAN, E., CHEN, F., ZHAO, H., FENG, J., XUE, J., HUANG, L., CHEN, J., SUN,

- Z., CHENG, Z., YIN, J., HE, Z., HUANG, L., WU, L., FEI, S., GU, S., JIANG, T., LI, T., CHEN, W., ZHOU, N., QIANG, N., LI, Q., HE, R., ZHANG, Y., LI, M., WANG, X., KASSEGNE, K., ZHU, Y., XIU, L., HU, Q., YIN, K., XIA, S., LI, S., WANG, Z., GUO, X., ZHANG, X. AND ZHOU, X. N. (2024). "How far has the globe gone in achieving One Health? Current evidence and policy implications based on global One Health index". *Science in One Health*, 3, e100064. Available at: https://doi.org/10.1016/j.soh.2024.100064 (Accessed: 30 June 2025).
- ZHOU, X. N., GUO, X. AND ZHANG, X. (eds.) (2025). *Global One Health Index Report 2022*. Singapore: Springer Nature Singapore. Available at: https://doi.org/10.1007/978-981-97-4824-2 (Accessed: 30 June 2025).
- ZOLO, D. (2000). *Cosmópolis: perspectiva y riesgos de un gobierno mundial*. Barcelona: Paidós Ibérica.
 - Normative and Juresprudencial References
- CESCR (Committee on Economic, Social and Cultural Rights) (2000). *General Comment No. 14: The Right to the Highest Attainable Standard of Health (Art. 12 of the ICESCR)*. UN Doc. E/C.12/2000/4 [online]. Available at: https://www.refworld.org/docid/4538838d0.html (Accessed: 02 July 2025).
- EUROPEAN UNION (2024). Directive (EU) 2024/1760 of the European Parliament and of the Council of 13 June 2024 on corporate sustainability due diligence and amending Directive (EU) 2019/1937 and Regulation (EU) 2023/2859 (Text with EEA relevance). Official Journal of the European Union. Available at: http://data.europa.eu/eli/dir/2024/1760/2025-04-17 (Accessed: 30 June 2025).
- UNITED NATIONS GENERAL ASSEMBLY (1948). *Universal Declaration of Human Rights*. Resolution 217 A (III), adopted 10 December 1948. Available at: https://www.un.org/en/about-us/universal-declaration-of-human-rights (Accessed: 12 July 2025).
- WORLD TRADE ORGANIZATION (1995). *Agreement on the Application of Sanitary and Phytosanitary Measures (SPS Agreement)*. Marrakesh Agreement Establishing the World Trade Organization, Annex 1A, 15 April 1994. Available at: https://www.wto.org/english/tratop e/sps e/spsagr e.htm (Accessed: 07 July 2025).
- WORLD TRADE ORGANIZATION (1995). *Understanding on Rules and Procedures Governing the Settlement of Disputes (Dispute Settlement Understanding, DSU)*. Marrakesh Agreement Establishing the World Trade Organization, Annex 2, 15 April 1994. Available at: https://www.wto.org/english/tratop_e/dispu_e/dsu_e.htm (Accessed: 07 July 2025).
- WORLD TRADE ORGANIZATION (1996). WT/DS26, European Communities Measures Concerning Meat and Meat Products, EC Hormones (US). Available at: https://www.wto.org/english/tratop_e/dispu_e/cases_e/ds26_e.htm (Accessed: 07 July 2025).

- WORLD TRADE ORGANIZATION (1996). WT/DS48, European Communities Measures Concerning Meat and Meat Products, EC Hormones (Canada). Available at: https://www.wto.org/english/tratop_e/dispu_e/cases_e/ds48_e.htm (Accessed: 07 July 2025).
- WORLDTRADE ORGANIZATION APPELLATE BODY (1998). *European Communities Measures concerning meat and meat products (Hormones)*, WT/DS26/AB/R and WT/DS48/AB/R, adopted 13 February 1998. Available at: https://www.wto.org/english/tratop e/dispu e/cases e/ds26 e.htm(Accessed: 10 July 2025).
- WORLD TRADE ORGANIZATION APPELLATE BODY (2002). *European Communities Trade description of sardines*, WT/DS231/AB/R, adopted 23 October 2002. Available at: https://www.wto.org/english/tratop_e/dispu_e/cases_e/ds231_e.htm (Accessed: 10 July 2025).
- WORLD TRADE ORGANIZATION (2003). WT/DS291, European Communities Measures Affecting the Approval and Marketing of Biotech Products, EC Approval and Marketing of Biotech Products. Available at: https://www.wto.org/english/tratop_e/dispu_e/cases_e/ds291_e.htm (Accessed: 07 July 2025).
- WORLD TRADE ORGANIZATION (2003). WT/DS292, European Communities Measures Affecting the Approval and Marketing of Biotech Products, EC Approval and Marketing of Biotech Products. Available at: https://www.wto.org/english/tratop e/dispu e/cases e/ds292 e.htm (Accessed: 07 July 2025).
- WORLD TRADE ORGANIZATION (2003). WT/DS293, European Communities Measures Affecting the Approval and Marketing of Biotech Products, EC Approval and Marketing of Biotech Products. Available at: https://www.wto.org/english/tratop e/dispu e/cases e/ds293 e.htm (Accessed: 07 July 2025).

Received: 16th July 2025 Accepted: 23rd September 2025

